Author’s response to reviews

Title: Comparative efficacy of Chinese medicine injections for treating acute cerebral infarction: A network meta-analysis of randomized controlled trials

Authors:

Shi Liu (1048823645@qq.com)
Jia-Rui Wu (exogamy@163.com)
Dan Zhang (2426394372@qq.com)
Kai-Huan Wang (wkhinda@163.com)
Bing Zhang (zhangbing6@263.net)
Xiao-Meng Zhang (fensexiaomao@sina.com)
Di Tan (tandi1993@qq.com)
Xiao-Jiao Duan (duanxiaojiaodgh@163.com)
Ying-Ying Cui (867422664@qq.com)
Xin-Kui Liu (1020172827@qq.com)

Version: 4 Date: 15 Mar 2018

Author’s response to reviews:

Dear Editorial Staff,

Thank you very much for processing our manuscript entitled “Comparative efficacy of Chinese medicine injections for treating acute cerebral infarction: A network meta-analysis of randomized controlled trials” (BCAM-D-17-01353)” reviewed in a timely manner.

We are really sorry for my carelessness which brings you much trouble. We have carefully corrected our submission. And we had resubmitted our paper without tracked changes, comments, highlights, strikethroughs or text in different colours. As for tables and figures, we already amended tables and depicted figures to 300 dpi.

If you have any queries, please don’t hesitate to contact me.

Thank you and best regards.
Yours sincerely,

Jia-Rui Wu
exogamy@163.com